Author’s response to reviews

Title: Weipiling ameliorates gastric precancerous lesions in Atp4a-/- mice

Authors:

Wei Liu (gzzyydxliuwei@126.com)
meng Zhao (88435811@qq.com)
liang Liu (448688667@qq.com)
feng Pan (gzphf@126.com)
zhu Lin (lizhulin26@yahoo.com)

Version: 4 Date: 10 Oct 2019

Author’s response to reviews:

October 11, 2019, Guangzhou

Dear Editor Arun K. Yadav,

Thank you very much for your kind letters dated on October 9, 2019, and sending us the information concerning our manuscript ‘Weipiling ameliorates gastric precancerous lesions in Atp4a-/- mice’. Enclosed please find out our revised version with some changes according to the consultant’s suggestions.

We greatly appreciate the consultant’s suggestions, and have revised the manuscript and would like to take the opportunity to point out as follows.

Editor Comments: Thanks again for attending all suggestions. However, please see that the scale bars on figures 2,3,4,5 are still not clearly readable. Please consult some published papers in BMC. The bar lines have to be prominent. May be you can double the length of scale bars and also the values of respective bar units, the line can also be little bit more in thickness. I feel this way all scale bars could be easily readable.

We greatly appreciate your comments, on the page 19-22, all the scale bars on photomicrographs of figures 2,3,4,5 have been modified to make them clearly readable.
We sincerely hope that the revised manuscript meets your approval and the approval of the reviewers, and that it is now acceptable for publication in your esteemed journal.

Thank you very much for your consideration of this paper.

Yours sincerely,

Hua-feng Pan, PhD, MD
Guangzhou University of Chinese Medicine,
Guangzhou, 510405, China